# **CSR FOR EMERGENCIES: THE TWO** CONCEPTS OF ACCOUNTABILITY

Christian Rainero<sup>\*</sup>, Giuseppe Modarelli<sup>\*\*</sup>

\* Department of Management, University of Turin, Italy \*\* Corresponding author, Department of Management, University of Turin, Italy Contact details: Department of Management, University of Turin, C.so Unione Sovietica 218-bis, 10136, Turin, Italy



## Abstract

How to cite this paper: Rainero, C., & Modarelli, G. (2020). CSR for emergencies: The two concepts of accountability. Corporate Ownership & Control, 18(1), 78-95.

http://doi.org/10.22495/cocv18i1art7

Copyright © 2020 The Authors

This work is licensed under a Creative Commons Attribution 4.0 International License (CC BY 4.0). https://creativecommons.org/licenses/by/ 4 0/

ISSN Online: 1810-3057 ISSN Print: 1727-9232

Received: 07.07.2020 Accepted: 14.09.2020

JEL Classification: L22, L26, M14, M31, M48 DOI: 10.22495/cocv18ilart7

This work tries to highlight the determinant role of CSR during periods characterized by non-linearity and to produce insights for further research on a dual perspective: 1) companies' decision-making on CSR implementation and promotional tool preference; 2) consumers' purchasing/consuming decision-making. So, the interrelated perspective, under the influencing contextual variable of the recent COVID-19 pandemic, would show a crucial role of CSR as an anti-crisis solution. The domain-based review, the direct observation mixed with field analysis survey-based on a sample of 208 respondents made possible the hermeneutical and inference activity on results according to the Situational Crisis Communication Theory and Goffman's view on the art of impression management, mixed with a persuasive approach characterizing advertising. The perception analysis, not directly operated ethnographically with companies, allowed to investigate CSR from different angles, permits the researchers to observe the exponential rise of advertising campaigns with a social scope and CSR promotional activities in accordance with the emergence of the two intrinsic concepts of accountability from an external point of view.

Keywords: CSR Advertising, COVID-19, Emergencies, Anti-Crisis, Accountability, Corporate Reputation and Image

individual contribution: Conceptualisation Authors' - G.M.; Methodology - C.R.; Resources - G.M.; Data Curation - G.M.; Formal Analysis - G.M.; Writing - C.R. and G.M.; Visualization -C.R. and G.M.; Supervision – C.R.

Declaration of conflicting interests: The Authors declare that there is no conflict of interest.

## **1. INTRODUCTION**

This research work aims to consider the concept of corporate social responsibility (hereinafter: CSR) as a relatively recent perspective that has emerged in the last two decades according to a renewed awareness and consciousness linked to globalized society, which for its intrinsic connotations constantly increases speed and the number of connections between places and people (Elbe, 2018). globalized society Todav the would show connotations and features of liquidity and discontinuity (Meyer, 2000; Drucker, 1969; Elbe, 2018) and an ever-growing development of exchanges with consequent environmental problems enhanced by a strong degradation of the soil, by the consumption of raw materials, pollution, and social repercussions related to new slavery at work. To make matters worse, the emergence of the recent COVID-19 pandemic. All these factors are undermining the model of a globalized society, inducing strong traits of discontinuity and fragmentation. As for the proposal of this document, according to the authors' point of view, the recent context characterized by a strong condition of non-linearity (due to the pandemic) which is the cause of a large health emergency worldwide, would call on companies to face the crisis and in parallel to contribute to the societal challenges. The authors' proposal would support the concept of CSR through a close relationship with the goal of companies to take responsibility for their impacts. Contrary to when this type of rational/charitable vision occurs in

VIRTUS 78

conditions of linearity, the role of CSR would find its alternative response also in a specific strategic approach for the improvement of external perceptions on the corporate image. For years, companies have preferred social/environmental/ sustainability reporting as an advertising tool, as it is aimed at involving a wide range of stakeholders with fewer resources invested. The recent COVID-19 pandemic would instead have a crucial role in changing this perspective, which could become more oriented to the consumer as the main stakeholder in the determination of business continuity. This would also frame a change in the methods and channels of communication, as well as the publicity of CSR data, in different cases, alternating the spirit and the real objective of promoting responsible activism, shifting the range of action from the real responsibility plan to the plan of the structuring of an impression engineering devoted to the implementation of anti-crisis tricks.

The growth of CSR would be considered as a preponderant part of the top-down decision-making process (Osemeke, Osemeke, & Okere, 2020), which leads (with greater probability) the big companies to deeply evaluate the impacts they produce on the social fabrics and on the tangible environment, requesting and introducing new opinions and perspectives.

For these reasons, the authors aim to propose an interesting interpretation of the recent reality relating to CSR and the emergency context in the sphere of investigation of corporate decision-making. Hence, the study would aim to strengthen the literature focusing on the power of CSR in (re-)enhancing the image and sales (indirectly) during critical events. On the other hand, the authors focus hermeneutic efforts on the emergency context as a determining factor in the influence of the corporate decision-making process and in the direction of the choice on CSR communication channels, from general stockholder-oriented tools and practices towards another more consumeroriented approach, through the use of social advertising or for social purposes (hereinafter S-ADV).

Therefore, the communication relating to CSR and the communication for advertising purposes would share the objective of maintaining and/or enhancing the return on the image. Unfortunately, little research focuses on the anti-crisis approach of the S-ADV, probably due to the difficulties in observing the emerging phenomena of the health crisis, such as the recent and most dramatic situation that occurred in the western territories after the Great Wars, due to the spread of the COVID-19 virus.

In summary, the research work conducted would seek to focus and underline the probable shift existing (during emergencies) in the choice of the communication channel relating to CSR (where possible) from traditional reporting to the advertising dimension, even more profitability-based, through the S-ADV. In this sense, the formation of a sort of anti-crisis device would be framed, through which companies could cope with emergency conditions under the aegis of direct and proactive communication, capable of producing a return on the corporate image in terms of responsibility, already widely attributed by the literature to the reasons underlying the choice to implement CSR interventions.

In literature, an unresolved paradox revolves around the role of CSR in behavioural consuming, if on one hand a strong intervention in charge of companies' awareness and accountability to societal challenges is required according to CSR activities reporting, contrarily Öberseder, Schlegelmilch, and Gruber (2011) and Mohr, Webb, and Harris (2001) note that consumers are relatively little interested in CSR according to their consuming and acquiring decision-making; in addition, Kolkailah, Aish, and El-Bassiouny (2012), declare that 57,9 of respondents are aware of socially responsible companies. By contrast, Arli and Lasmono (2010) declare that when consumers have to buy similar products at the same price level and quality, CSR would be a strong determinant in decision-making, because consumers would prefer to buy from firms with socially responsible reputation. From this perspective, the research proposed would investigate, through field analysis and qualitative data inferred by the questionnaires (No. 208), adding information on the paradox aforementioned (under the variable the emergency condition and usual-linear of purchasing activity).

Subsequently to the domain-based review (Paul & Criado, 2020), the study attempts to demonstrate the crucial role of CSR, according to the shifting choice to S-ADV in protecting companies during critical/non-linear events, as the recent observed COVID-19 virus spread.

The authors also attempt to provide insights on the theme through a field analysis based on direct observation of increased S-ADV production, verifying the assumption made with a control survey and perception analysis.

The core of the research would be based on the shift of channel to communicate CSR during crisis/emergencies (from shareholders to consumers). It is well known in the literature that companies are used to structure annual reports in publicizing and communicate voluntarily (when not obliged) to disclose non-financial information to a wide range of stakeholders (Mihalache, 2011).

Especially among managerial literature, the tendency would be to match CSR promotion with social/environmental/sustainability/integrated reporting as predominant tools to make public the data referable to the scope of the report (Hess, 2008).

Conversely advertising appears a field of study more developed in marketing, psychology, and other behavioural sciences. The authors try to use the emergency condition to bridge, in a multidisciplinary way, two ambits that appeared separately during the last decades. In fact, to study advertising from the business point of view appears a little unusual (more strategical perspective), but the recent COVID-19 pandemic offers to social researchers an unrepeatable observation field, opening a window to investigate CSR and S-ADV as strictly linked solutions for companies and a first step to cope with the economic crisis.

Through this view and the interpretative frame of the arts of impression management and theory of presentation of self-provided by E. Goffman (1956), CSR could enhance image and reputation, furnishing sometimes helping messages and strong influence on humans (Bryant & Zillman, 2002; Bandura,1965) through TVs also boosting and stimulating consumers' decision-making, orienting their choices.

VIRTUS

According to Dillard and Vinnari (2019) and Brown and Dillard (2015), CSR reporting would be strengthened by the dialogical power of ADV.

In this sense, the complexity that calls companies to face constant changes and crisis makes it necessary a strong approach to the constitution of task forces internal to the company with the aim to manage emergencies related to the capability in overturning negative repercussions to renewed opportunities.

The authors investigate the two concepts of CSR and the shifting activities in decision-making during non-linear context (crisis/emergency) from a traditional CSR communication tool to another one as an active crisis-response strategy.

As observed by Sanclemente-Téllez (2017), in recent years CSR has affected several specific sectors and segments, increasing interest among managers, scholars, and practitioners, stimulating attempts to incorporate the concept and the related activities into a wider context. In this wake, the authors attempt to match the CSR to S-ADV favoring insights and further development on the importance to generate value true a kind of marketing solution, especially during an emergency condition.

Theoretical perspectives and conceptual interpretations have been proposed, making possible inferences and interconnected views.

The formal structure of the articles follows these points: Section 2 shows the research design and method; Sections 3, 4, and 5 respectively refer to an interrelated literature review on CSR, CSR from reporting to ADV and CSR during emergencies; Section 6 explains the referring context and interpretative frame; Section 7 shows the results critically discussing them and Section 8 concludes.

## 2. LITERATURE REVIEW: CORPORATE SOCIAL RESPONSIBILITY

The contemporary conceptions related to corporate governance are converging on the fact that corporate culture would be a crucial resource for the company and this perspective, integrated with the capabilities to satisfy stakeholders' needs, would make the evolution of the business in strict connection with the concept of perdurance (Zappa, 1957; Puddu, 2010). That is the globalized society cannot ignore the moral approach to entrepreneurial activities, legitimating ethics in economics in contrast with the human being as a "rational maximizer subject" shaped by the classical view (Posner, 1997). This vision already noted by Smith (1863) has been restored by more recent authors as Scott (2015).

The well-known contrast between Friedman's stockholders perspective, for which CSR would be functional and instrumental to the main scope of businesses to increase profits (Friedman, 2007) and the more ethics-integrated view by Freeman (1984) and Evan and Freeman (1988) continues to produce debates (Grove, Clouse, & Xu, 2020), according to the importance to embrace societal challenges in charge of entrepreneurial activities and common goods (Frey, 2005; Zamagni, 2011). Authors as Devinney (2009) concentrate the efforts on the binomial contrast between the existence or not of a true CSR (or the ideal concept deserved to the non-profit sector). The contemporary condition of emergency

and the extraordinary non-linear context would renew the question: is the homo *"homo oeconomicus"* really disappeared?

Indirectly the research follows this guide-line through the interpretative activity approached by the observation of increased S-ADV strictly connected to CSR during COVID-19 spread. It would be necessary to say that the definition of CSR appears not specifically univocal (Öberseder et al., 2011), complex and in several cases unclear, but centered on recursive pillars. In fact, it refers to the responsible actions that companies implement to balance their impacts on society, improving community well-being thanks to discretionary activities and resource involvement in contributing social challenges (Kotler & Lee, to 2005) environmental, ethical, and social benefits (Vaaland, Heide, & Grønhaug, 2008; Bhattacharya & Sen, 2004).

CSR is broadly defined as an organisation's social obligation that extends beyond the scope of the firm and requirements of the law, to further a social cause (Bhattacharya & Sen, 2004). CSR has long been on the agendas of governments. Despite it would be still an evanescent concept that could mean different things according to different stakeholders, specific pillars, several as aforementioned, contribute to shaping the lines of its substantial meaning. According to OECD (2001) in a globalized society, companies play a crucial role beyond job creation and growth, integrating a contribution to sustainable development (Brown, Hanson, Liverman, & Merideth, 1987).

In a globalized environment, awareness of social actions has significantly grown<sup>1</sup>. On this base, it would be possible to observe a need required by companies to be responsible (also strategically) operating cuts to the "rational-irrationality" in boosting only profits at the expense of other social and environmental assets and resources. So, CSR would appear in the business scenario when "companies decide voluntarily to contribute to a better society" (European Commission, 2011, as cited in Mihalache, 2011, p. 159).

The entrepreneurial environment nowadays is called to come ever-growing uncertainty and pressure, and the aim to continue operating on the market would enhance the need to transfer a strong reputation, often through CSR activities.

In this sense, market pressure would be shifted on the moral and strategic plan. Rethinking to what aforementioned, for which the authors declare the presence of two perspectives on CSR and the existing separation in the literature between profit-orientation (stockholder view - strategical power of CSR) and the moral direction of entrepreneurial activity (ideal of societal challenges integration through CSR in business governance orientation). These views have been widely discussed in the literature, but nowadays, also under the lens of COVID-19 spread, in abstract would prevail the idea of businesses moved by the sense of obligations not only to their shareholders but also to the whole society (Boulouta & Pitelis, 2014; Backhouse & Wickham, 2020). At this point, a question would emerge spontaneously: what is the role of companies and more in general organizations into the actual society? Do they live as

<sup>&</sup>lt;sup>1</sup> https://www.fridaysforfutureitalia.it/

monads, self-referential (Janssen, Charalabidis, & Zuiderwijk, 2012), or they born from society, for society, to satisfy needs before making profits?

Under this profile, the economic choice in consuming takes in place the main actor – the consumer. Consumers appear to be increasingly oriented to responsibility, primarily in consuming behaviours, environmental and societal, and also to the responsibility through which companies operate (Chamhuri & Batt, 2015; Lagerkvist, Okello, & Karanja, 2015). The emerging attention poses new challenges for companies that need to communicate their activity to the external, also posing questions on the strategic choice of tools to use, according to the subjects involved.

Reflecting on the actual economic development, it would be possible to say that globalization and ICTs are changing and accelerating transformation processes covering several ambits of life (Chizh & Gavrylenko, 2016), by contrast, the sense of uncertainty and the loss of reputation is becoming a wide spectrum focal point in business management.

A well-perceived role of companies in society would mitigate the disastrous impact of industries on the environment. In this sense, not only shareholders, but a wide range of stakeholders are interested in companies' reputation, and corporations are called to necessarily emphasize the CSR-related image of the firm to acquire institutional legitimacy (Khan, Lew, & Park, 2015).

When image and reputation are requested, marketing strategies need attention. On one hand, traditional tools, born to make public data on CSR, would be social/environmental/sustainability reporting, but also marketing campaigns per se (more expensive) would be useful strategies to present S-ADV, the deserving ambit of CSR under the light of humans behaviours actually conflicting with the devastating consequences of "homo oeconomicus" (Martin-Sardesai & Guthrie, 2019).

From a strategical perspective, CSR would be recognizable as a crucial operational direction to orient actions to tangible and perceived benefits for organisational and social outcomes, depicting a kind of win-win solution (Kerr, Johnston, & Beatson, 2008). These two perspectives during a crisis could tend to merge together, creating a valid dichotomy able to match different interests and needs.

The knowledge about the benefits deriving from CSR appears advanced during the last years. In this sense, several authors underline business responsible behaviour as an influencing factor to be perceived socially responsible and to be positively evaluated by consumers with positive repercussions on product sales (Bhattacharya & Sen, 2004), without prejudice of personal values propension (Lee & Cho, 2018). With the aim to be perceived in a positive manner by stakeholders, enhancing the reputation, companies are implementing actions devoted to CSR integration in their business models. These efforts would have been analyzed by several under different perspectives authors and interpretative lens, responding to needs, and emerging social challenges. Communication models and approaches of social responsibility activities are becoming crucial points in companies' agendas, mostly during a crisis, when reputation and image would be the main critical factor of success. Authors as Ken (2016) investigate the abandonment of CSR during the financial crisis. Kim and Choi (2018)

concentrate the efforts on the post-crisis effect of communication. Idowu, Vertigans, and Burlea (2017) focus the attention on CSR in times of crisis with a cross-country approach. Others as Uddin, Siddiqui, and Islam (2018) investigate on CSR disclosure in the banking system. Fiorani, Meneguzzo, and Jannelli (2012) provide an interpretative view on CSR 2.0, based on a proactive approach to sustainability between global markets and crisis-management. Many ambits have been investigated, contrarily, it would be possible to say that CSR has been studied fragmentarily from an integrative perspective based on reporting and in the ambit of marketing (Sanclemente-Téllez, 2017) under non-linear conditions (COVID-19).

## 2.1. CSR from reporting to advertising

For several years and still, nowadays, academics, practitioners, and managers have discussed the existing links between business and society, according to their interdependencies with the context of CSR and stakeholder theory (Brown & Forster, 2013; Carroll, 1979; Freeman, 1984).

The classic concept of "rational maximizer" related to human-being and translated in profit maker (in business terms), concentrating efforts only on shareholders' interests, would considerably change once emerged stakeholders' needs and societal challenges in a globalized world.

If globalization would be considered as a fracture in social life able to change balances between internal and external factors, both among the members of communities and among companies, so the latter will find solutions to cope with emerging complexity (OECD, 2001; Koppenjan & Klijn, 2004).

Reflecting the more recent definition of CSR, it would consider a renewed approach in doing entrepreneurial activities, linked to the manner through which companies manage their operations (through sustainable, economical, social, and environmental attention) recognizing interests to a wide-spectrum audience (from shareholders to consumers) considering a sustainable development (Sanclemente-Téllez, 2017).

According to what aforementioned, CSR would mean that businesses, in addition to maximizing shareholder value, must act in a manner that benefits society, also contributing to the welfare of society and the environment (Ganti, 2019).

The Cambridge dictionary defines CSR as "the idea that a company should be interested in and willing to help society and the environmentas well as be concerned about the products and profits its makes"<sup>n</sup>.

Another on line dictionary definition of CSR focuses the attention on "the obligation that companies have to be good citizens of their community and of the world, actively considering their impact on employees, consumers and the environment"<sup>3</sup>. According to what reported by European Commission Green Paper, "CSR involves a company's voluntary integration of social and environmental concerns regarding its business operations and relationships with representatives" (European Commission, 2001).

<sup>&</sup>lt;sup>2</sup> https://dictionary.cambridge.org/us/dictionary/english/corporate-social-

responsibility <sup>3</sup> https://www.yourdictionary.com/corporate-social-responsibility

Moreover, "CSR is indicated as to reinforce the development of new markets and opportunities for growth and innovation since it requires enterprises to carefully follow changing social expectations and ensure consumer trust through adopting new sustainable business models" (European Commission, 2011).

Considering the question that the economic health of a society is generally held to be a universal benefit, any activity that contributes to growth should be a socially responsible activity. By contrast, CSR is a multilevel concept and should be applied with care and deep consideration under its several facets (Preston, 2005).

Someone creates a framework (Kerr et al., 2008), someone concentrates the efforts on religiosity and CSR efforts (Mazereeuw-van der Duijn Schouten, Graafland, & Kaptein, 2014), others investigate on the CSR influence on loyalty (Iglesias, Markovic, Bagherzadeh, & Singh, 2020), but little attention has been deserved in the literature to more specific intervention in academic ambit, that concentrate the efforts on the role of CSR during the crisis and the tool preference to communicate these data (Ham & Kim, 2019).

It appears important to say that corporate financial profitability would constitute a key factor influencing corporate social/environmental/ sustainability initiatives and reporting/ communicating activities (Fuhrmann, 2019).

When companies choose to implement CSR, they also refer to the context in which they operate and to the referring period. In this sense, they also need to choose the communication tool to promote CSR activities, consciously operating a decisionmaking process (Figure 1). Switching from reporting to ADV, probably when an element of rupture in social life affects the sphere of normality, would favor the rise of the concept of relevance and which information convey to the external (Simons, 2009), not all the information would be made public, and during crises, a deeper choice on the most suitable tool would be necessary, according to the relevance of the information to be disclosed directly.

Despite the positive consequences that can result, as well as for integrated reporting, barriers to implementation may also be present for social advertising (Dumay, Bernardi, Guthrie, & La Torre, 2017). In this direction, however, the recent emergency COVID-19 offers an unrepeatable opportunity to evaluate to what extent and in which way companies prefer to contribute to the generation of value from a dual profile: 1) of social responsibility and 2) strategical influence in consuming behaviours through the preference of a communication tool than another.

Social/environmental/sustainability reporting and integrated reporting are documentally a more complete pattern than ADV spots related to CSR. By contrast, marketing campaigns are able in few seconds to express how the firm is and how its CSR is devoted. ADV would appear an easier way to ensure comprehensibility and intelligibility or readability, because of its dialogical power in regards to consumers, favoring word of mouth (WOM) positive influence (Pang & Lee, 2008) and for these reasons, could be preferred by companies as an anti-crisis solution to boost sales during emergencies.

VIRTUS

Also, the purpose of social reporting/ environmental/sustainability and integrating reporting would be to communicate in a simple manner how the companies are involved to generate value for social/environmental aspects, according to wide-spectrum concrete repercussions on society.

In this sense, the concept of materiality or relevance (Krahel & Titera, 2015) would acquire importance referring to the CSR activities and information sharing to the external. Materiality would be a multifaceted concept that permeates the whole accounting field, more in general considering the need to provide relevant information to users of corporate reporting (Bernstein, 1967).

The concepts of materiality/relevance and readability/comprehensibility/intelligibility would be strictly connected, and research in business, management, and accounting are focusing the attention on these aspects related to the complexity of information inherent in the documents and reports in this ambit.

The tension and conflictual relation emerged according to the possibility to match stakeholders' needs for the useful and conscious decision on an informed base, make crucial the decision on appropriate communication tools (Lamprecht & Guetterman, 2019).

In this sense, the value concerning information relevance would be enhanced according to corporate responsibility reporting activity, embracing the idea of providing non-financial information to a wide range of stakeholders.

The readability and the relevance of summarized information involved would play a crucial role in shareholders/stockholders' decision-making, but also on the consumers' decision-making side (de Klerk & de Villiers, 2012).

The slogan "Doing well by doing good", used by Stroup and Newbert (1987), is becoming part of large companies' vision, that decide to implement CSR initiatives using marketing strategies to make public the data and emphasizing their relevance (Mihalache, 2011).

That approach would profoundly modify the role of companies in society, shaping them from (in several cases) irrational-resource-consumers to social-responsible-producers. But the rationality behind that approach would be based on the possible image enhancement that in the long-run would produce benefits centered on rational self-interest, while various stakeholders (also consumers) decide to support the company because of its responsibility (Chahal & Sharma, 2006).

In fact, Handelman and Arnold (1999) demonstrate the positive connection between CSR and companies' performance. In this sense, companies seem pleased to integrate social or environmental proposition in their marketing strategies to orient the business direction, according to the appropriate tool/channel choice to make CSR data public.

In recent decades, according to the governmental orientation to guarantee a sustainable development for the future generations (Dresner, 2008), corporate social/environmental/sustainability disclosure has acquired a central role in managerial, practitioners, and academics debates, especially in accounting issues (Gray, 2006).

Several debates occurred in the literature about the possibility to present separate disclosure of non-financial information about CSR, that would constitute an easier way to communicate data, by contrast, the disclosure of a single document would increase the risks of information overload (Kleijnen, Lee, & Wetzels, 2009) and mitigate these conflicts a solution emerged thanks to the integrated reporting tool (Fuhrmann, 2019).

Investors and stakeholders rely on high-quality integrated reports to obtain social/environmental and ethical information for conscious decision-making (Ellis, 1989; du Toit, van Zyl, & Schütte, 2017).

On the base of materiality/relevance readability/comprehensibility/intelligibility, the aim of integrated reporting would be to present the information in a concise manner, linking different aspects and dimensions according to the proposal to focus attention on value creation (Macias & Farfan-Lievano, 2017).

So, the integrated report shares with ADV and S-ADV, the dialogical power, and communicative direction (Dillard & Vinnari, 2019; Brown & Dillard, 2015).

Into this perspective, the reporting activity on CSR initiatives would be more complex than it seems, shaping a multifaceted view able to reflect, on one hand, a firm response to a wide range of stakeholders' expectations, on the other one, a strategical asset in promoting CSR initiatives, also allocating a consistent amount of budgets to that (Yuan, Lu, Tian, & Yu, 2020). By contrast, implementing social responsibility would be facilitated through standards and common codes of conduct (OECD, 2001).

Integrated reporting would share with the ADV communication tool a strategical view for growth (Macias & Farfan-Lievano, 2017). Massa, Farneti, and Scappini (2015) discuss on the SMEs objectives to start reporting activities on CSR, especially due to establish and enhance a positive consequence on the reputation side.

De Villiers, Hsiao, and Maroun (2017) individuated determinant factors of this kind of reporting including size, profitability, growth potential, leverage financial performance, position, ownership structure, managerial attitude and motivations, organizational culture, corporate governance, non-financial performance, disclosure costs. Also, they focus the attention on the consequences of this kind of reporting adoption including long-term social and environmental sustainability, financial market stability, value relevance, information environment, legitimacy, accountability, image enhancement, stakeholder engagement, mitigation of litigation risk.

So, these attributes would be referable to the CSR ADV solutions (Figure 1).

Previous studies have found that CSR operations have several advantages in times of crisis. In addition, it is assumed that CSR efforts contribute to generating a favorable relationship between a company and its stakeholders.

Iglesias et al. (2020) underline the power of CSR in terms of image creation/restoring, according to the 2016 Global RepTrak, CSR would be a crucial antecedent of corporate reputation, linked to better financial performance (Boulouta & Pitelis, 2014).

VIRTUS

According to what aforementioned, the dispute on an economistic view (Friedman, 2007) and more humans-related one (Freeman, 1984; Evan & Freeman, 1988) rediscovers value if related to CSR, enhancing the concept of competitiveness and strategies connected to enhance companies' positive perception. In this wake, Sanclemente-Téllez (2017) systematically collects definitions of CSR and marketing, matching their meanings and attributions.

### 2.2. Advertising for CSR during emergencies: An anti-crisis consumer-oriented approach

In literature, great interconnections between the ethical responsibility of companies and the instruments used to be or to be perceived as responsible have been provided by Pantani et al. (2016).

They shape the idea related to the ethical responsibility of the firms in terms of activities deserved to the avoidance of social and environmental damage even without earning money.

So, this kind of altruistic behaviour, could be included in philanthropic and/or humanitarian interventions as a "re-pay solution" to the whole society. By contrast, over time, this kind of ideal responsibility, started to be considered under the strategical point of view, involving opportunistic behaviours for reputation increasing and profit-orientation in the long-run, also manipulating companies' external perception and legitimizing operational procedures.

Companies usually use these kinds of "cosmetic strategies" since the dawn of ADV, just thinking of "*carosello*" (carousel).

In the first episode aired, an ADV spot was provided by a preeminent international company, actually operating in the energy sector, contributing in favor of road safety, teaching drive techniques through TVs.

S-ADV is undoubtedly reflected in its early days, and today it is called upon to play a dual role: to promote adequate behaviours and a strategical anti-crisis solution for companies in contexts of non-linearity.

Often people are used to see responsible and sometimes antithetical promotional practices, for example, campaigns of "drinking responsibly" related to the sale of alcohol.

Main authors and researchers in legitimacy theory ambit suggest that businesses attempt to follow and achieve a wide approval of stakeholders, related to their operational (Guthrie & Parker, 1989; Bebbington, Larrinaga-González, & Moneva-Abadía, 2008).

Businesses that are able to realize high profits, tend to be under pressure and under the lens of public judice, increasing visibility. By contrast, that perspective is configured with a dual profile, on the one hand, a growth in reputation, on the other, its management.

In this sense, making interventions in favor of social/environmental ambits would be a strategical and crucial channel to move legitimacy toward the company (Naser & Hassan, 2013).

Therefore, the information provided with reference to the financial and non-financial, would explain better the company position, favoring broader market evaluation. Private and institutional investors, shareholders, and analysts would consider a clearer frame based on different points of view (de Klerk & de Villiers, 2012).

The sustainability disclosure and social/environmental reporting, under the light of integrated reports, would stimulate interest on which tool would be better to show non-financial information towards different stakeholders (Massa et al., 2019).

Referring to this kind of activity, important insights would emerge aligning the social/environmental/sustainability reporting tool to the S-ADV instrument, in terms of marketing promotional campaigns.

Framing of the marketing concept, functional to the discourse, would be necessary. Individuating the more recent definitions of marketing, the authors try to dispel the crucial meanings of the concept.

According to a definition provided in two versions among 2007 and 2013 by the AMA's Committee, marketing would the activity, set of institutions, and processes oriented to create, communicate, deliver and offer, producing value for consumers, clients, partners, and the whole society.

The Merriam-Webster Dictionary defines marketing as *"the process or technique of promoting, selling, and distributing a product or service"*<sup>4</sup>.

Another online dictionary defines marketing as "the act and process of buying and selling"<sup>5</sup>.

The Collins Dictionary provides another definition reported as follows: "marketing is the organization of the sale of a product, for example, deciding on its price, the areas it should be supplied to, and how it should be advertised".

Alexandra Twin (2020) answers the question "what is marketing?" highlighting the main characters of the concept that and declaring that "marketing refers to activities a company undertakes to promote the buying or selling of a product or service. Marketing includes advertising, selling, and delivering products to consumers or other businesses".

If marketing would be strictly related to advertising, a definition would be crucial.

The Merriam-Webster Dictionary defines advertising as *"the action of calling something to the attention of the public especially by paid announcements"*<sup>8</sup>. At this stage it would be possible to hermeneutically include both activities (marketing and advertising), referring to the more summarized and holistic definition of marketing provided by Cambridge Dictionary, it would be "a job that *involves encouraging people to buy a product or service"*<sup>9</sup>.

Clear and important connections between advertising and marketing, would shape the lines of inferential possibilities. Advertising and its relation with consumers would be a pillar in a capitalist system for boosting future expenditure (Preston, 2005).

So, the influence that TVs and ADV would have on consuming-behaviours is largely recognized in the literature (Keller, 1987). Opinions circulated concerning the bad effects of advertising on children (Preston, 2005). In complex, divergent points of view emerged among who is in favor of ADV and who is a detractor.

There are certainly opened questions about the responsibility that the ADV industry has toward the society (Preston, 2005) and for specific segments of the population, according to its strong power in influencing behaviours. According to the definition of marketing provided by the Collins Dictionary, it would be possible to shape the line of ethical aspects, incorporated in marketing campaigns. In fact, marketing is becoming to incorporate "ethical elements, such as the public interest, and environmental concern into the planning of business strategy"<sup>10</sup>.

To criticize marketing would be very simple especially under the lens of return-benefits that it produce including CSR promotional aspects both with traditional reporting (Martin-Sardesai & Guthrie, 2019) and ADV campaigns.

So, CSR could become a cosmetic treatment, an economic strategy to boost sales, and an anti-crisis trick. Several authors as Martin and Schouten (2014), McKenzie-Mohr (2011), and Gordon, Carrigan, and Hastings (2011) decided to pursue the objective of studying sustainable perspectives in this ambit.

Theoretically, the S-ADV solution (including ethical aspects) and the social/environmental or sustainability reporting (including non-financial information), share the ideal to promote something good for society in terms of responsibility and ethics in economics and business. But what happens during times of crisis and emergency?

During crises, time would be crucial, and bad decisions about timing and approach models of communicating can lead to serious losses and even failure.

actions would Therefore strategical be but external determinants necessary, and influencing factors, such as the recent health emergency (COVID-19), could guide (depending on the structural situation of the company - e.g., its profitability, which also influences CSR interventions) the choice towards a channel of social/environmental activities promotion to the stakeholders.

While the documents relating to social/environmental/sustainability reporting would traditionally be used in contexts of normality and more referable to shareholders and indirectly in a simplified manner to stakeholders in general (including consumers); contexts characterized by non-linearity, such as emergencies, would take on specific characteristics relating to the strategic choice of communication in the ambit of social responsibility, orienting more towards consumer-oriented channels (S-ADV). Recent COVID-19

<sup>&</sup>lt;sup>4</sup> https://www.merriam-webster.com/dictionary/marketing

<sup>&</sup>lt;sup>5</sup> https://www.yourdictionary.com/marketing

<sup>&</sup>lt;sup>6</sup> https://www.collinsdictionary.com/it/dizionario/inglese/marketing

<sup>&</sup>lt;sup>7</sup> https://www.investopedia.com/terms/m/marketing.asp <sup>8</sup> https://www.merriam-webster.com/dictionary/advertising

<sup>&</sup>lt;sup>9</sup> https://dictionary.cambridge.org/it/dizionario/inglese/marketing

<sup>&</sup>lt;sup>10</sup> https://www.collinsdictionary.com/dictionary/english/corporate-socialresponsibility

emergency made more frequent and visible the incremental number of S-ADVon TVs.

Probably many companies have been driven by the same reason that moves them to produce social/environmental/sustainability reports: for promotional purposes to advertising their business and responsible behaviours, guaranteeing legitimacy and return of image and reputation.

Anti-crisis management is impossible without marketing management (Chizh & Gavrylenko, 2016).

In recent years, researchers posed attention to transversal perspectives related to CSR suggesting that it can favor positive reactions in companies' stakeholders enhancing commitment and identification (Hildebrand, Sen, & Bhattacharya, 2011).

In literature, a great interest emerged on investigating the impact of CSR on product and company evaluation.

So, commitment, consumer identification with the company, consumers' perceptions of companies' motives for implementing and promoting CSR, awareness in what CSR is, would be determinants for a positive evaluation of the business, enhancing legitimacy and modifying consuming decision-making and product attractiveness (Öberseder et al., 2011). In the terms of anti-crisis management, it would be required to choose the better tool and the better strategy to promote CSR and boost return-benefits for the company (Chizh & Gavrylenko, 2016), also ensuring perdurance on time and space of the business activity (Puddu, 2010).

That strategical view on tool choice for promoting CSR interventions would shift the ideal of CSR, involving an ethical-oriented view into an anticrisis trick, strictly related to the marketing activity.

This frame would show a probable dark-side of CSR that could be named corporate social performance (CSP) (Kerr et al., 2008) describing the processes and the outcomes enacted by CSR.

Mahajan (2011) states that many factors influence CSR interventions, including globalization, governmental and inter-governmental bodies awareness in sustainability practices, advances in communication technology, the need for improving social responsibility against breaches to ethics, citizens' awareness, and aim in social/environmental caring and regulatory initiatives.

The implementation of social/environmental or sustainability reporting would be enhanced by the need to promote CSR interventions in normal conditions, deserving a little attention or almost additional to more expensive S-ADV campaigns as the primary channel of communication to inform or influence stakeholders. COVID-19 pandemics would make it more difficult to advertise social actions through traditional reporting tools, therefore social advertising proliferates during emergencies/crises.

For this reason, the authors would add "emergency condition" to the CSR list provided by Mahajan (2011), not only as a determinant but also as a facilitator to move the promotional channel for CSR from traditional reporting to consumer-oriented S-ADV.

A consequence in the last few years, and the strong numerosity in the recent COVID-19 period, would be the decision made by several brands to start implementing CSR promotional campaigns thanks to S-ADV on a dual perspective: 1) inserting it in the core of the business strategy as a renewed sense of consciousness and 2) inserting it in the core of the crisis-management strategical approach.

Someone really considers the genuinity of CSR interventions without promoting them. Some others prefer to invest money and a large part of budgets in promotional activities related to CSR (Iglesias et al., 2020). So, the recent COVID-19 pandemics would offer a strong opportunity to shift CSR from the ideal and genuine plan to a strategical asset in promoting adequate behaviours expecting a positive repercussion in terms of reputation, image, loyalty, profits, and legitimacy.

The activity devoted to edulcorate the ideal of CSR to strategically boost sales during the crisis, would appear enhanced in the recent emergency period.

From this perspective, it would be plausible that when consumers tend to perceive brands or are led to believe the brand as socially responsible, they would be also more favorable to spend money for their products, according to the effect of trust and loyalty creation (Iglesias et al., 2020).

Consumers could be considered as one of the main influencing actors of the business, including the fact that their choices would determine the preferences on goods and services, especially in crisis periods, when reductions of expenses occur. So, the anti-crisis management needs to consider that for further development and perdurance of the company (Chizh & Gavrylenko, 2016).

For this reason, companies decided to use CSR interventions and their promotion to make the brand more palatable to the stakeholders (Lindgreen & Swaen, 2010), initially, thanks to less expensive documents internally made as to the social/ environmental/sustainability reports are, rarely accompanied by S-ADV (Kerr et al., 2008), and subsequently, especially in crisis/emergency period, the strategical orientation (profitability-based) would move to the enhanced use of CSR promotional campaigns, consumer-oriented (Figure 1).

VIRTUS 85



Figure 1. Company CSR decision-making process and preferred promotional tool integration

*Source: Our processing and further elaboration from Wood (1991).* 

#### 2.3. Referring context and interpretative frame

To better frame the interpretative vision and the context in which the observation occurs, a specific specification would be necessary. Initially, the authors decided to outline the boundaries of the term "crisis"; in fact, a crisis would generally be attributable to the definition of a sudden and unexpected event with negative impacts.

With the aim of significantly reducing the negative impact of the crises, for some years drastic changes have taken place in terms of efforts that many companies have decided to support in the field of CSR, according to the more general objective of combining moral imperatives/social environmental, or indirectly increasing competitiveness, reputation, legitimacy on the market, etc. An increasing number of companies have decided to start investing money in CSR activities and to promote them on different channels.

Traditionally, social/environmental or sustainability reports have been used (tools more oriented towards shareholders and interested parties in general) and only S-ADV (more expensive) campaigns were implemented in a residual way to communicate a specific socially responsible activity, showing in this way activism concerning social environmental and sustainability issues, with a focus on results more specifically oriented towards consumer-engagement and readability/intelligibility.

The pandemic period from COVID-19 and the consequent health emergency worldwide would show a greater production of S-ADV attributable to several large companies.

Management literature suggests that one of the reasons why companies invest in terms of money and human resource efforts in their CSR operations would focus on trusting the positive repercussions that these efforts have on the corporate image and against possible losses and damages caused by emergencies in times of crisis (Ponomarenko & Marinina, 2017).

This evidence would question the reasons behind the implementation of CSR between companies, which would not necessarily be linked to altruistic feelings (Miras-Rodríguez, Escobar-Pérez, & Galán-González, 2015).

Several researchers such as Coombs and Holladay (2006) suggest that a previous positive reputation can generate the so-called "halo effect", capable of protecting companies when the crisis could adversely affect the image causing losses.

According to their point of view, where companies appear to use CSR in crisis management practice to avoid negative repercussions, the authors

NTER PRESS VIRTUS

include this study in the more managerial perspective provided by Coombs and Holladay (2002), and provided in the more sociological framework: the arts of impression management and self-presentation theory (Goffman, 1956).

As for the former perspective (Coombs & Holladay, 2002; Ham & Kim, 2019), thanks to CSR-based crisis response strategies, companies could protect the corporate image, reducing the risks associated with the state of crisis and emergency.

In Situational Crisis Communication Theory (Coombs & Holladay, 2002), in close synergy with a self-presentation strategy (Goffman, 1956) and a persuasive approach (Ham & Kim, 2019), the companies could proactively respond to critical situations and to the negative consequences deriving from them, preventing or facing them. The recent pandemic spread from COVID-19 has paralyzed daily operations around the world.

The epidemic first, which then became a pandemic, caused by an unknown virus, would have involved countries, individuals, organizations, and companies to react to potential pathogens and the consequent disruptive negative repercussions related to the health and economic crisis.

The world has always suffered from man-made and natural epidemics and disasters, preparing reactions and planning solutions to deal with the unexpected collapse at various management levels.

A solution adopted by companies and observed in the period to which explicit reference was made previously, would be the greater production of S-ADV, which could in fact be traced back to the ambit of communication theory in crisis situations, following in the wake of the well-known advantages deriving from CSR interventions in terms of image return.

Reflecting on the growing interest in reputation management and crisis management over the past decade, the authors try to provide a definition of reputation, according to Goffman (1956) regarded as the art of impression. In this sense, according to the managerial perspective, the latter would be identifiable as the management of impressions, which focuses attention on the self-presentation/ presentation of the self, influencing the external evaluation. Indeed, Fombrun and van Riel (1997) define reputation as a cognitive evaluation of the audience.

If companies were considered to be actors in global markets, under the interpretative lens of Goffman's dramaturgical performance, they would be "actors acting with expressive responsibility" (Goffman, 1956, p. 132).

Considering this approach in terms of managing impressions, the authors can formulate the identifying features of the crucial role played by the promotional activity of CSR under normal conditions (linearity) and its role reinforced during emergencies through the crisis management approach.

Drama actors are known to use techniques to save the show. By widening the perspective to business organizations (companies, businesses, etc.), the latter could be considered as agents of social life in business contexts, and therefore able to avoid interruptions and unexpected events, making it possible to continue the business through space and time even when the crisis and external interference affect their stability (Puddu, 2010). Goffman (1956) states that the actors do not care about the moral issue of making these standards, "but about the amoral issue of engineering a convincing impression that these standards are being realized" (p. 162).

In this perspective, it would be possible to interpret the implementation of CSR as an impression activity (through traditional reporting) even during periods characterized by linear conditions. Consequently, translating the involvement of this activity into periods characterized by non-linearity (crisis/emergencies etc.) (through S-ADV), would constitute a considerable opportunity for the exit or in any case the reduction of the negativities produced by the crisis, in terms of increasing profits through the orientation of purchase/consumption behaviours (without prejudice to the possible existence of ideal CSR promotion actions that go beyond opportunistic behaviours).

With this objective, traditional reporting (aimed at publicizing the results and social/environmental/ sustainability activities promoted by companies) would show a dialogic power, which would be strengthened in its strategic dimension by the persuasive approach provided by the ADV (Dillard & Vinnari, 2019; Brown & Dillard, 2015).

#### **3. RESEARCH DESIGN AND METHOD**

Social initiatives would include not only the traditional practice of corporate philanthropy but can also involve multiple varieties of forms related to corporate support for communities (Khan et al., 2015). The proposition of this research tries to explore the role of CSR as a failure-risk-reducing determinant through promotional tools as S-ADV.

The study proposed aims to investigate the emergency variable as an antecedent of shifting determinant in companies' decision-making on the preferable tool to promote CSR interventions during the crisis, also providing a frame on consuming behaviours during emergencies and on CSR as a double faces concept considered both an ideology and a strategical asset, under the lens of processes of anti-crisis management, able to generate favorable conditions for the companies' survival redefining models of efficiencies into a context characterized by non-linearity.

For these reasons, and due to the multiple aspects that affect the investigation, the purpose of this study would focus its objectives in developing insights to the readers, involving them toward the interpretative lens of an anti-crisis management perspective in observing of a phenomenon in rapid growth and change (the recent COVID-19).

In this frame Mihalache (2011) furnishes connections between CSR and marketing, also suggesting that CSR and its direct/indirect impacts on society could be read through the choice of the more suitable tool for publicizing its interventions and data.

In fact, in this paper, the authors examine the relationship between emergency conditions and channels to promote CSR activities, involving into the paradigm of how, also the paradigm of why.

The authors started the research according to the direct observation, which produced the first insight to move the investigation in a deeper direction. Following the aims explicated, the authors

VIRTUS 87

try to investigate the ambit through the observation mediated by ADV hubs on the internet (www.spotandweb.it) intersected with a gap identification and a field analysis based on attitudes and perceptions extrapolated thanks to the help of questionnaires (Krosnick & Presser, 2009) would be useful to triangulate data and better frame the ambit of investigation and answer RQs.

If on one hand the direct observation on TVs and the mediated observation on the main ADV hubs on the internet, suggest enhanced attention to S-ADV during COVID-19 pandemics, the empirical research conducted among 208 respondents, would be necessary to extrapolate qualitative and quantitative data to validate or refute assumptions deriving from literature.

In that way it has been possible to shape the lines of the research, explicitly linking CSR marketing strategies under several perspectives, focusing the investigation on specific RQs, concerning the whole society, companies, and consumers behavioural choices.

The authors, also attempt to provide an interesting study offering exploratory contributions to the existing literature on CSR and promotional tools for its activities (intended as marketing strategies *lato sensu*), favoring an unrepeatable context of observation thanks to the emergence of COVID-19 emergency all around the world, for single persons, families, social and economic institutions and actors.

At first, the authors provide a structured literature review in the following paragraphs, based on a multidisciplinary integration of CSR among accounting and marketing, through decision-making and consuming behavioural approaches, also influenced by transversal psychological research.

The transversality characteristics of CSR could appear fragmented, contrarily that would be a strong opportunity to open new paradigms of interpretation.

One hand, literature has been observed that a very small segment of consumers considers CSR during the purchasing/consuming decision-making in normal conditions (Mohr et al., 2001).

According to that view, questions would emerge spontaneously:

*RQI:* If in normal conditions consumers do not consider CSR much, what happens in emergency conditions?

*RQ2:* During the emergency does consumer social responsibility-based choice increase (even if induced by the marketing strategy)?

RQ3: Can CSR be considered as real responsibility or as a business promotion tool to influence purchasing behaviour during emergencies?

*RQ4: Which CSR promotional tool do companies prefer during crises (where possible to implement)?* 

*RQ5:* Can the advertising tool with a social purpose (*S*-*ADV*) be a self-referencing anti-crisis cosmetic trick to retain and increase profits during emergencies?

It is well known in methodology literature that the qualitative approach cannot provide standard measures, but can provide additional and transversal information to quantitative analysis. So, the authors try to match both perspectives through a quali-quantitative approach, probably preferable to answer the RQs at this stage of the research, favoring insights for deeper further analysis, more specific and focused on restricted variables. In this sense, to ensure quite a robustness of the data, the authors follow well-established criteria as transferability and confirmability (Bunz, Casulli, Jones, & Bausch, 2017) able to generalize and reproduce the results.

The idea at the base of the research that considers under several conditions CSR and crisis communication able to share the same intents leads the authors to build a specific research design based on a domain-based review (Paul & Criado, 2020) and following the guidelines of the RQs declared. In this sense, the authors use the ideal and strategical perspective of CSR and the marketing lens on the promotional tool/channel to use during a period characterized by non-linearity (emergency/crisis).

So, CSR through traditional reporting or S-ADV share the aim of maintaining companies' reputation, providing an anti-crisis solution to boost sales, but little research deserved attention (because of little opportunities occurred until recently, due to COVID-19 spread) on how the anti-crisis managerial perspective would affect CSR interventions promotional tool choice.

The considerations of the literature that concentrate efforts especially on the reporting activity side would create a gap in strategic assessment oriented to the decision-making on which tool would be preferable to use during a crisis, with the aim to communicate CSR activities to consumers, also providing a quite marketing strategy to boost sales, orienting consuming behaviours and probably manipulating the CSR ideal moving it from true accountability to anti-crisis trick.

The authors initiate domain-based investigations (Paul & Criado, 2020) on the main internet databases of academic literature, specifically identified for management ambit: Scopus (filtered both by "academic journal" and "business, management, accounting" criteria), EconLit (filtered by "academic journal") and Business Source Ultimate (filtered by "academic journal").

The criteria used to limit and restrict the research ambit (by titles, abstracts, and keywords), would be the following (Figure 2): "CSR" or "Corporate Social Responsibility" and "Strategy", and "Crisis" and "Anti-crisis".

This view would ensure a strong segmentation and profiling of a gap identification. In fact, CSR in general among the three databases would show the wider academic literary production, that once restricted to the ambit of "Strategy" related to the "Crisis" context, shows No. 111 results on Scopus, No. 136 results on Business Source Ultimate, and No. 39 on EconLit.

To accurately trace CSR back to strategic orientation in a context of crisis as a resolutive and/or avoidance case, another criterion ("Anti-crisis") has been added. From this preliminary investigation, a single contribution on Scopus resulted (Ponomarenko & Marinina, 2017) as reported in Figure 2.

In this direction, the authors have been able to identify a gap in the non-direct possibility of obtaining academic literary results on CSR as an anti-crisis solution.

VIRTUS



Figure 2. Gap identification procedure

Source: Our elaboration.

Through the direct observation on TVs and web-based research on one of the main Italian hub that deserves a section of the site to the new ADV spots, reviewing them.

The authors considered the period between 1st of March 2020 and 20th of May 2020, discovering that about 52% of the spots reported by the web-site (www.spotandweb.it) during the aforementioned period, was social-scope-related. With the aim to validate the data and add value to the direct observation triangulating results, the authors deserved attention to perceptions provided through a field analysis survey-based by a sample of 208 respondents (voluntary-based randomized selection) on 271 involved, with a response rate about 77%.

The sample composition is based on 27% males and 73% females, 22,60% between 30-44 years old, 1,44% between 18-29 years old, and 75,96% between 45-64 years old. The questionnaire, structured on 9 affirmative points, allowed respondents to graduate answers on a five-level Likert scale, that the authors restricted to better show the results, incorporating 1-2 (lower accordance), 3 (medium accordance), 4-5 (higher accordance).

One of the points included in the survey has been extrapolated, because of the transversal importance, used as a verification factor of direct observation.

In fact, as reported in Figure 3, 28,37% of the respondents declare a perceived growth in S-ADV spots during the COVID-19 equal to a range of +25-35%; 32,69% of respondents involved declare a growth of about +35-50%, finally, 29,33% of respondents note the growth in S-ADV during the aforementioned period greater than 50%. These values would validate the direct observation operated by the authors.

Figure 3. Perception of increased S-ADV spots



## 4. FIELD OF ANALYSIS RESULTS

Going further companies' ideal and moral interventions during societal crisis/emergency response, researchers try to integrate SCCT with the persuasion knowledge model (Ham & Kim, 2019).

So, this would have made it possible to produce inspirational insights capable of bringing the authors, under the interpretative model before explained, to go beyond the traditional view of government intervention in the economy (Borre & Viegas, 1995) and state help to the companies when external unexpected causes of major force occur, investigating on pro-active solutions able to integrate strategical decision-making on the more suitable tool to use in promoting CSR.

In this way, the CSR would be used as a strategical expedient to model own image, benefiting of positive impact, well known, that CSR interventions have on consumers' perceptions, also involving the influence power that characterizes the ADV campaigns (although more expensive than traditional reporting).

So, the authors, presenting the data extrapolated by the analysis of questionnaires, after direct observation of increased production of S-ADV, would show the connected impact of the crisis on behavioural intentions both on consumers and CSR promotional tool preference.

The interpretative analysis suggests that S-ADV would be considered as an anti-crisis strategical solution.

The authors, as mentioned in the section, deserved to research design and method, involving

a sample of 208 respondents, structuring 9 affirmative sentences (except the one extrapolated before in Figure 3), allowing respondents to graduate answers on a five-level Likert scale, restricted to 3-aggregated-points with the aim to better show the data, incorporating 1-2 (lower accordance), 3 (medium accordance), 4-5 (higher accordance).

The affirmative sentences submitted to the 208 respondents involved are reported as follows:

1. The COVID-19 emergency has increased volunteering and donation actions;

2. The emergency has increased the production of advertising for social purposes to encourage the adoption of appropriate behaviours to the situation;

3. I am more oriented to buy products from companies that show social responsibility;

4. I am willing to spend more on goods produced by companies that do charities;

5. I am willing to spend more on goods produced by companies that act with respect for the planet;

6. The criteria in the last 3 points (social responsibility, charities, respect for the planet) are decisive in my usual choice of purchase or consumption;

7. In the current emergency condition, I prefer to purchase products promoted through actions and promotional spots of social responsibility;

8. The emergency has increased the production of advertising for social purposes to guide consumer choices.

Figure 4 shows the results emerging from questionnaires.



#### Figure 4. Data analysis

Source: Our elaboration.

The points 1, 2, and 8 have been structured to know the respondents' perceptions on the referring context and the emergency as an influencing variable; the points 3 and 6 have been structured to know the orientation of the sample involved; the points 4, 5, and 7 have been structured to know the propensity of the sample involved in purchasing/consuming behaviours and deducing the companies' strategical decision-making in shifting CSR promotion from traditional reporting to S-ADV as an anti-crisis tool.

Point 1 would show a strong agreement (73,63%) on the increasing observation of volunteering and donation actions, reflecting a great existing media influence on promoting and reiterate these kinds of activities. The point 2 would reflect the emergency condition as an influencing factor in exceptional increased production of advertising for social purposes, that the sample involved recognizes with a percentage of agreement about 80,77%, perceiving the promotional activity as able to encourage the adoption of appropriate behaviours to

<u>VIRTUS</u>

the situation. The point 8, by contrast, would deserve an important consideration on the fact that 68,27% of the respondents consider that the emergency condition has increased the production of advertising for social purposes to guide consumer choices. Interpreting these data it would be possible to suggest that a great quota of the sample would be aware of the fact that, if on one hand, S-ADV would encourage appropriate behaviours, on the other one, would enlarge profits basin through influence on purchasing/consuming decision-making.

Point 3 would show a powerful orientation (67,31%) of the sample involved in purchasing goods produced by companies able to show social responsibility. In this sense, while several authors as Mohr et al. (2001) underline that only a very small segment of consumers consider CSR as a determinant factor during the purchasing or consuming decision-making in normal conditions, the data presented would follow another point of view, through which it would be possible to interpret corporate responsible behaviours as influencing factors to be evaluated and positively perceived 2004). bv consumers (Bhattacharya & Sen, Demonstrating the positive connection between CSR and companies' performance (Handelman & Arnold, 1999). The view would be enhanced by point 6, through which 69,71% of the respondents consider as determinants of usual purchasing/consuming decision-making the following criteria: social responsibility, charities, respect for the planet.

The points 4 and 5 would demonstrate that the sample involved, would be willing to spend more for goods produced by companies that act responsibly to the planet (81,73%) and to the people who live social disadvantages (61,06%). That orientation would be empowered by the current emergency condition (point 7). In fact, a large part of the sample involved would prefer to purchase goods promoted through actions and spots of social responsibility (59,62%).

## 5. DISCUSSION OF RESULTS: THE TWO CONCEPTS OF ACCOUNTABILITY

Thanks to the hermeneutical approach used by the authors to analyze the data, the RQs that defined the investigation design would find answers.

In fact, the response to RQ1 would reflect, conditions of non-linearity (emergency, etc.), in a strong preference to consider CSR. Contrary to what is reported in a small segment of the literature (Mohr et al., 2001), not only a limited portion of consumers would consider CSR during purchasing operations and in consumer choices, but social responsibility, charitable and beneficial actions, as well as respect for the planet, would be decisive in the daily purchasing activities of a large part of consumers. Referring to RQ2, after analyzing the data, it could be considered that the awareness present on communication via S-ADV as an influencing factor, would increase the choice towards social responsibility of consumers orientations, fulfilled through actions advertised by the manufacturing companies.

As regards RQ3, RQ4, and RQ5, CSR could be considered through a two-pronged perspective: 1) a perspective of real responsibility and 2) a perspective focused on the type of behavioural influence, especially during emergencies. Indeed, systematically observed and validated by as triangulating data derived from consumer perception, companies would prefer S-ADV as a promotional tool to show CSR interventions. constituting a crucial resource on which to invest in terms of anti-crisis management. This second face or dark side of CSR would accompany the general ideal underlying the same, favoring the performance of companies and modeling the forms of the S-ADV as a self-referencing anti-crisis cosmetic expedient to preserve and increase profits during emergencies.

## 6. CONCLUSION

Several Scholars argue that corporate social responsibility (CSR) is considered by corporations as strategic for the business because of its contribution to financial performance (Sharp & Zaidman, 2010).

The literature about CSR could be divided into two viewpoints: 1) the more business-oriented (Friedman, 2007) and 2) the more stakeholderoriented (Freeman, 1984; Evan & Freeman, 1988). According to the latter, it would be possible to consider "purchasing always a moral act and not simply an economic choice"<sup>11</sup>. During emergencies this sentiment would be enhanced. For this reason, the study focuses the attention on a little bit neglected area in literature, related to CSR and anticrisis strategical approach, showing a kind of darkside behind a wonderful ideal.

So, through direct observation and survey-based field analysis, the results would show how companies, on one hand, would try to acquire legitimacy implementing social responsibility, on the other hand, try to acquire benefits through S-ADV.

The emergency period also would constitute а moment of strategical decision-making on investment in CSR and related choice on more adequate promotional tool. Several authors underline the abandon of CSR during financial crisis (Ken, 2016). In this sense, a double way would guide the choice on promotional tool, depicting a crucial role in anti-crisis management. Abandon or enhancement? The insights provided by the analysis of data, suggest that S-ADV would be preferred during crisis for its direct capability to communicate (Coombs & Holladay, 2015) through a dialogical power to the consumer, that even if present in traditional social/environmental/sustainability reports, would be enhanced by media influence, becoming an essential asset for boosting sales in periods characterized by non-linearity.

Research has established connections between CSR and crises, on the company side and purchasing/consuming decision-making on consumer side. Also, the authors have shown how the COVID-19 pandemic ensures an unreproducible playground of study about wide phenomena, as the S-ADV appeared during the period of its major spread. That would have helped researchers to understand how promotional communication and CSR-related actions can be used to protect companies during emergencies as recent pandemics.

CSR efforts could be considered as a specific strategical asset during crisis, and the two concepts of accountability would take shape, shifting from

 $<sup>^{11}</sup> https://www.columban.org.au/media-and-publications/articles/features/2019/acrath-campaign-purchasing-is-always-a-moral-act.$ 

societal challenges ideals to "cosmetic tricks" on accountability through strategical decision-making on promotional tool to use.

Indeed, as reported by Hildebrand et al. (2011), CSR would seem to be an almost perfect vehicle for corporate marketing and within emergency conditions, this would provide a preferential motivation for the implementation of the S-ADV, without prejudice to traditional reporting.

In detail, however, it would be somewhat complex to establish when the CSR and S-ADV interventions are structurally true in the ideal and when these are strongly oriented by strategic motivations without the aid of observations made with the help of corporate insiders. In this sense, the strengthening of the relationship between ideal and integrated thinking, capable of making the business transparent and also showing efforts in other social/environmental areas, on the one hand, could favor external communication and the solidity of reputation without manipulation, even through S-ADV and/or traditional reporting; on the other, the logic of "*Homo oeconomicus*" and of the "rational maximizer of one's goals", in times of emergency, could come back into vogue.

If on one hand, as the authors try to show through this explorative study, during emergencies CSR could assume a double perspective in terms of strategical decision-making investments for anti-crisis management, the S-ADV could boost influence sentiments in consuming decision-making, implementing (for more profitable companies) ADV campaigns, shifting the traditional CSR reporting tools into alternative communication strategies.

At a theoretical level, one of the contributions of the paper would contrast with what is reported in a small segment of the literature for which only a limited portion of consumers would consider CSR during decision-making in consuming habits. In addition, the main contribution of the paper would reflect an unreproducible opportunity to frame companies' behaviours during the COVID-19 pandemic spread, according to their objective to satisfy needs and persist over time. So, the context of non-linearity would offer a great field of study especially considering CSR as an anti-crisis managerial tool. In this sense, two ideals behind the CSR and accountability would be clearly visible, one more oriented to collective benefits, the other one more oriented to companies' survival.

Therefore, social communication would tend to resemble more a strategy aimed at camouflaging rather than transmitting the true corporate identity, capturing an instrumentally opaque legitimacy. This limit, extremely linked to the qualitative aspects of the investigation, could be filled in subsequent research, through a detailed analysis of the individual S-ADV spots in light of the historical background of the company's social responsibility actions.

While CSR is intrinsically linked to profitability and investment capacity, many smaller companies, although responsible agents, would remain in the shadows, at a disadvantage, with less or almost totally denied access to promotional communication strategies and take-over engagement possibilities with consumers. For these reasons, the real extent of the phenomena would be difficult to represent and in the current phase, the study would provide embryonic insights, encouraging multidisciplinary efforts, aware that in the recent emergency period, even companies not strictly dedicated to CSR have been able to implement promotional activities through S-ADV following opportunistic behaviours in the wake of anti-crisis strategies. The difficulty in observing phenomena in rapid growth and evolution, and the complexity in shaping the lines of their dynamism, could reflect limitations in methodological approach to the design of the research especially in social sciences field. In this case, the perception analysis, not directly operated with companies, allowed to investigate CSR from different angles, permitted the researchers to observe the exponential rise of advertising campaigns with social scope and CSR promotional activities, by contrast, it was not possible to establish when the CSR and S-ADV interventions were structurally true in the ideal and when these were strongly oriented by strategic motivations, providing only insights. To fill this limitation it would have been possible (but really complex to implement during the lockdown) to orient the study with the help of internal observators, through an ethnographic approach made with the help of corporate insiders. Further research could implement the study in this direction providing a quite generalizable referring frame on the theme, but considering that the period of interest (early stages of COVID-19 pandemic) may no longer be reproducible and therefore a similar subsequent period will have to be taken into account.

#### REFERENCES

- 1. Arli, D. I., & Lasmono, H. K. (2010). Consumers' perception of corporate social responsibility in a developing country. *International Journal of Consumer Studies*, *34*(1), 46-51. https://doi.org/10.1111/j.1470-6431.2009.00824.x
- 2. Backhouse, K., & Wickham, M. (2020). Corporate governance, boards of directors and corporate social responsibility: The Australian context. *Corporate Ownership & Control, 17*(4), 60-71. http://doi.org/10.22495/cocv17i4art5
- 3. Bandura, A. (1965). Influence of models' reinforcement contingencies on the acquisition of imitative response. *Journal of Personality and Social Psychology, 1*(6), 589-595. https://doi.org/10.1037/h0022070
- 4. Bauman, Z. (2000). *Liquid modernity* (1st ed.). Cambridge, UK: Polity Press.
- 5. Bebbington, J., Larrinaga-González, C., & Moneva-Abadía, J. M. (2008). Legitimating reputation/the reputation of legitimacy theory. *Accounting, Auditing & Accountability Journal, 21*(3), 371-374. https://doi.org/10.1108/09513570810863969
- 6. Bernstein, L. A. (1967). The concept of materiality. *The Accounting Review*, 42(1), 86-95.
- 7. Bhattacharya, C., & Sen, S. (2004). Doing better at doing good: When, why, and how, consumers respond to corporate social initiatives. *California Management Review*, *47*(1), 9-25. https://doi.org/10.2307/41166284
- 8. Borre, O., & Viegas, J. M. (1995). Government intervention in the economy. In O. Borre & E. Scarbrough (Eds.), *The scope of government* (Beliefs in government, Vol. 3, pp. 234-279). Oxford, England: Oxford University Press.

VIRTUS

- 9. Boulouta, I., & Pitelis, C. N. (2014). Who needs CSR? The impact of corporate social responsibility on national competitiveness. *Journal of Business Ethics*, *119*, 349-364. https://doi.org/10.1007/s10551-013-1633-2
- 10. Brown, B. J., Hanson, M. E., Liverman, D. M., & Merideth, R. W. (1987). Global sustainability: Toward definition. *Environmental Management*, *11*(6), 713-719. https://doi.org/10.1007/BF01867238
- 11. Brown, J. A., & Forster, W. R. (2013). CSR and stakeholder theory: A tale of Adam Smith. *Journal of Business Ethics*, *112*, 301-312. https://doi.org/10.1007/s10551-012-1251-4
- Brown, J., & Dillard, J. (2015). Dialogic accountings for stakeholders: On opening up and closing down participatory governance. *Journal of Management Studies, 52*(7), 961-985. https://doi.org/10.1111/joms.12153
  Bryant, J., & Zillman, D. (Eds.). (2002). *Media effects: Advances in theory and research* (2nd ed., LEA's
- Bryant, J., & Zhinnan, D. (Eds.). (2002). Media effects: Advances in theory and research (2nd ed., EEA's communication series). https://doi.org/10.4324/9781410602428
  Bunz, T., Casulli, L., Jones, M. V., & Bausch, A. (2017). The dynamics of experiential learning: Microprocesses
- 14. Bunz, 1., Casuili, L., Jones, M. V., & Bausch, A. (2017). The dynamics of experiential learning: Microprocesses and adaptation in a professional service INV. *International Business Review*, *26*(2), 225-238. https://doi.org/10.1016/j.ibusrev.2016.07.001
- 15. Carroll, A. B. (1979). A three-dimensional conceptual model of corporate social performance. *Academy of Management Review, 4*(4), 497-505. https://doi.org/10.5465/amr.1979.4498296
- 16. Chahal, H., & Sharma, R. D. (2006). Implications of corporate social responsibility on marketing performance: A conceptual framework. *Journal of Services Research, 6*(1), 5-6. Retrieved from https://www.researchgate.net/publication/285854144\_Implications\_of\_corporate\_social\_responsibility\_on\_mar keting\_performance\_A\_conceptual\_framework
- 17. Chamhuri, N., & Batt, P. J. (2015). Consumer perceptions of food quality in Malaysia. *British Food Journal, 117*(3), 1168-1187. https://doi.org/10.1108/BFJ-08-2013-0235
- 18. Chizh, V., & Gavrylenko, V. (2016). Models of crisis management in a changing environment. *Scientific Journal of Polonia University*, *18*(3), 105-112. https://doi.org/10.23856/1812
- 19. Coombs, W. T., & Holladay, S. J. (2002). Helping crisis managers protect reputational assets: Initial tests of the situational crisis communication theory. *Management Communication Quarterly*, *16*(2), 165-186. https://doi.org/10.1108/CCIJ-10-2013-0078
- 20. Coombs, W. T., & Holladay, S. J. (2006). Unpacking the halo effect: Reputation and crisis management. *Journal of Communication Management*, *10*(2), 123-137. https://doi.org/10.1108/13632540610664698
- 21. Coombs, W. T., & Holladay, S. J. (2015). CSR as crisis risk: Expanding how we conceptualize the relationship. *Corporate Communications: An International Journal, 20*(2), 144-162. https://doi.org/10.1108/CCIJ-10-2013-0078
- 22. de Klerk, M., & de Villiers, C. (2012). The value relevance of corporate responsibility reporting: South African evidence. *Meditari Accountancy Research, 20*(1), 21-38. http://dx.doi.org/10.1108/10222521211234200
- 23. de Villiers, C., Hsiao, P. C. K., & Maroun, W. (2017). Developing conceptual model of influences around integrated reporting, new insights and directions for future research. *Meditari Accountancy Research*, *25*(4), 450-460. https://doi.org/10.1108/MEDAR-07-2017-0183
- 24. Devinney, T. (2009). Is the socially responsible corporation a myth? The good, the bad, and the ugly of corporate social responsibility. *The Academy of Management Perspectives, 23*(2), 44-56. https://doi.org/10.5465/amp.2009.39985540
- 25. Dillard, J., & Vinnari, E. (2019). Critical dialogical accountability: From accounting-based accountability to accountability-based accounting. *Critical Perspectives on Accounting*, 62, 16-38. https://doi.org/10.1016/j.cpa.2018.10.003
- 26. Drucker, P. F. (1969). *The age of discontinuity: Guidelines to our changing society*. New York, NY: Routledge.
- 27. du Toit, E., van Zyl, R., & Schütte, G. (2017). Integrated reporting by South African companies: A case study. *Meditari Accountancy Research*, *25*(4), 654-674. https://doi.org/10.1108/MEDAR-03-2016-0052
- 28. Dumay, J., Bernardi, C., Guthrie, J., & La Torre, M. (2017). Barriers to implementing the international integrated reporting framework: A contemporary academic perspective. *Meditari Accountancy Research*, *25*(4), 461-480. https://doi.org/10.1108/MEDAR-05-2017-0150
- 29. Elbe, S. (2018). *Pandemics, pills and politics: Governing global health security*. Baltimore, MD: Johns Hopkins University Press.
- 30. European Commission. (2001). *Green paper: Promoting a European framework for corporate social responsibility*. Retrieved from https://ec.europa.eu/commission/presscorner/detail/en/DOC\_01\_9
- 31. European Commission. (2011). Communication from the Commission to the European parliament, the Council, the European Economic and Social Committee and the Committee of the Regions: A renewed EU strategy 2011-14 for corporate social responsibility. Retrieved from http://eurlex.europa.eu/LexUriServ/LexUriServ.do?uri =COM:2011:0681:FIN:en:PDF
- 32. Evan, W. M., & Freeman, R. E. (1988). *A stakeholder theory of the modern corporation: Kantian capitalism*. New York, NY: University Press.
- 33. Fiorani, G., Meneguzzo, M., & Jannelli, R. (2012). CSR 2.0 proattiva e sostenibile: Tra mercati globali e gestione della crisi. Milano, Italia: Egea.
- 34. Fombrun, C., & van Riel, C. (1997). The reputational landscape. *Corporate Reputation Review*, *1*, 5-13. https://doi.org/10.1057/palgrave.crr.1540024
- 35. Freeman, R. E. (1984). Strategic management: A stakeholder approach. Boston, MA: Pitman.
- 36. Frey, B. S. (2005). Non-solo per denaro: Le motivazioni disinteressate dell'agire economico. Milan, Italy: Mondadori Bruno.
- Friedman, M. (2007). The social responsibility of business is to increase its profits. In W. C. Zimmerli, M. Holzinger, & K. Richter (Eds.), *Corporate ethics and corporate governance* (pp. 173-178). https://doi.org/10.1007/978-3-540-70818-6\_14
- 38. Fuhrmann, S. (2019). A multi-theoretical approach on drivers of integrated reporting Uniting firm-level and country-level associations. *Meditari Accountancy Research, 28*(1), 168-205. https://doi.org/10.1108/MEDAR-12-2018-0412
- 39. Ganti, A. (2019). *What is social responsibility?* Retrieved from https://www.investopedia.com/terms/s /socialresponsibility.asp

VIRTUS

- 40. Goffman, E. (1956). *The presentation of self in everyday life* (University of Edinburgh Social Sciences Research Center, Monograph No. 2). Retrieved from https://monoskop.org/images/1/19/Goffman\_Erving\_The\_Presentation \_of\_Self\_in\_Everyday\_Life.pdf
- 41. Gordon, R., Carrigan, M., & Hastings, G. (2011). A framework for sustainable marketing. *Marketing Theory*, *11*(2), 143-163. https://doi.org/10.1177/1470593111403218
- 42. Gray, R. (2006). Social, environmental and sustainability reporting and organisational value creation? *Accounting, Auditing & Accountability Journal, 19*(6), 793-819. https://doi.org/10.1108/09513570610709872
- 43. Grove, H., Clouse, M., & Xu, T. (2020). Long-term stockholder and stakeholder value and corporate governance implications. *Corporate Law & Governance Review, 2*(1), 18-26. http://doi.org/10.22495/clgrv2i1p2
- 44. Guthrie, J., & Parker, L. D. (1989). Corporate social reporting: A rebuttal of legitimacy theory. *Accounting and Business Research*, *19*(76), 343-352. https://doi.org/10.1080/00014788.1989.9728863
- 45. Ham, C., & Kim, J. (2019). The role of CSR in crises: Integration of situational crisis communication theory and the persuasion knowledge model. *Journal of Business Ethics*, *158*, 353-372. https://doi.org/10.1007/s10551-017-3706-0
- 46. Handelman, J. M., & Arnold, S. J. (1999). The role of marketing actions with a social dimension: Appeals to the institutional environment. *Journal of Marketing*, *63*(3), 33-48. https://doi.org/10.1177/002224299906300303
- 47. Hess, D. (2008). The three pillars of corporate social reporting as new governance regulation: Disclosure, dialogue, and development. *Business Ethics Quarterly, 18*(4), 447-482. https://doi.org/10.5840/beq200818434
- 48. Hildebrand, D., Sen, S., & Bhattacharya, C. B. (2011). Corporate social responsibility: A corporate marketing perspective. *European Journal of Marketing*, *45*(9/10), 1353-1364. http://dx.doi.org/10.1108/03090561111151790
- 49. Idowu, S. O., Vertigans, S., & Burlea, A. S. (Eds.). (2017). *Corporate social responsibility in times of crisis: Practices and cases from Europe*. https://doi.org/10.1007/978-3-319-52839-7
- 50. Iglesias, O., Markovic, S., Bagherzadeh, M., & Singh, J. J. (2020). Co-creation: A key link between corporate social responsibility, customer trust, and customer loyalty. *Journal of Business Ethics*, *163*, 151-166. https://doi.org/10.1007/s10551-018-4015-y
- 51. Janssen, M., Charalabidis, Y., & Zuiderwijk, A. (2012). Benefits, adoption barriers and myths of open data and open government. *Information Systems Management, 29*(4), 258-268. https://doi.org/10.1080/10580530.2012.716740
- Keller, K. L. (1987). Memory factors in advertising: The effect of advertising retrieval cues on brand evaluations. *Journal of Consumer Research*, 14(3), 316-333. https://doi.org/10.1086/209116
- 53. Ken, W. (2016). Crisis narratives and the abandonment of CSR during the financial crisis: Notes from systems integrated. *Social and Environmental Accountability Journal, 36*(3), 188-202. https://doi.org/10.1080/0969160X.2016.1246375
- 54. Kerr, G., Johnston, K., & Beatson, A. (2008). A framework of corporate social responsibility for advertising accountability: The case of Australian government advertising campaign. *Journal of Marketing Communications*, *14*(2), 155-169. https://doi.org/10.1080/13527260701858539
- 55. Khan, Z., Lew, Y. K., & Park, B. I. (2015). Institutional legitimacy and norms-based CSR marketing practices: Insights from MNCs operating in a developing economy. *International Marketing Review*, *32*(5), 463-491. https://doi.org/10.1108/IMR-01-2014-0017
- 56. Kim, S., & Choi, S. M. (2018). Congruence effects in post-crisis communication: The mediating role of attribution of corporate motives. *Journal of Business Ethics*, *153*(2), 447-463. https://doi.org/10.1007/s10551-016-3425-y
- 57. Kleijnen, M., Lee, N., & Wetzels, M. (2009). An exploration of consumer resistance to innovation and its antecedents. *Journal of Economic Psychology*, *30*(3), 344-357. https://doi.org/10.1016/j.joep.2009.02.004
- Kolkailah, S. K., Aish, E. A., & El-Bassiouny, N. (2012). The impact of corporate social responsibility initiatives on consumers' behavioural intentions in the Egyptian market. *International Journal of Consumer Studies, 36*(4), 369-384. https://doi.org/10.1111/j.1470-6431.2011.01070.x
- 59. Koppenjan, J., & Klijn, E.-H. (2004). *Managing uncertainties in networks: Public private controversies* (1st ed.). https://doi.org/10.4324/9780203643457
- 60. Kotler, P., & Lee, N. (2005). *Corporate social responsibility: Doing the most good for your company and your cause.* Hoboken, New Jersey: John Wiley & Sons, Inc.
- 61. Krahel, J. P., & Titera, W. R. (2015). Consequences of big data and formalization on accounting and auditing standards. *Accounting Horizons*, *29*(2), 409-422. https://doi.org/10.2308/acch-51065
- Krosnick, J. A., & Presser, S. (2009). Question and questionnaire design. In J. D. Wright & P. V. Marsder (Eds.), Handbook of survey research (2nd ed., pp. 1-81). Retrieved from https://www.betterevaluation.org/en /resources/question-and-questionnaire-design
- 63. Lagerkvist, C. J., Okello, J. J., & Karanja, N. (2015). Consumers' mental model of food safety for fresh vegetables in Nairobi: A field experiment using the Zaltman Metaphor Elicitation Technique. *British Food Journal*, *117*(1), 22-36. https://doi.org/10.1108/BFJ-09-2013-0280
- 64. Lamprecht, C., & Guetterman, T. C. (2019). Mixed methods in accounting: A field based analysis. *Meditari Accountancy Research*, *27*(6), 921-938. https://doi.org/10.1108/MEDAR-11-2018-0403
- 65. Lee, J., & Cho, M. (2018). New insights into socially responsible consumers: The role of personal values. *International Journal of Consumer Studies*, *43*(2), 123-133. https://doi.org/10.1111/ijcs.12491
- 66. Lindgreen, A., & Swaen, V. (2010). Corporate social responsibility. *International Journal of Management Reviews*, *12*(1), 1-7. https://doi.org/10.1111/j.1468-2370.2009.00277.x
- 67. Macias, H. A., & Farfan-Lievano, A. (2017). Integrated reporting as a strategy for firm growth: Multiple case study in Colombia. *Meditari Accountancy Research*, *25*(4), 605-628. https://doi.org/10.1108/MEDAR-11-2016-0099
- 68. Mahajan, P. (2011). Corporate social responsibility: A new wave of corporate governance. Advance online publication. https://doi.org/10.2139/ssrn.1840601
- Martin, D. M., & Schouten, J. W. (2014). The answer is sustainable marketing, when the question is: What can we do? *Recherche et Applications En Marketing (English Edition), 29*(3), 107-109. https://doi.org/10.1177/2051570714540497
   Martin-Sardesai, A., & Guthrie, J. (2019). Social report innovation: Evidence from a major Italian bank 2007-
- 2012. Meditari Accountancy Research, 28(1), 72-88. https://doi.org/10.1108/MEDAR-10-2018-0383
- 71. Massa, L., Farneti, F., & Scappini, B. (2015). Developing a sustainability report in a small to medium enterprise: Process and consequences. *Meditari Accountancy Research*, *23*(1), 62-91. http://dx.doi.org/10.1108/MEDAR-02-2014-0030

<u>VIRTUS</u>

- 72. Mazereeuw-van der Duijn Schouten, C., Graafland, J., & Kaptein, M. (2014). Religiosity, CSR attitudes, and CSR behavior: An empirical study of executives' religiosity and CSR. *Journal of Business Ethics*, *123*(3), 437-459. https://doi.org/10.1007/s10551-013-1847-3
- 73. McKenzie-Mohr, D. (2011). *Fostering sustainable behavior: An introduction to community-based social marketing* (3rd ed.). Gabriola Island, Canada: New society publishers.
- 74. Meyer, J. W. (2000). Globalization: Sources and effects on national states and societies. *International Sociology*, *15*(2), 233-248. https://doi.org/10.1177/0268580900015002006
- 75. Mihalache, S.-S. (2011). CSR A marketing tool? *Annals of Spiru Haret University, Economic Series, 11*(1), 159-164. Retrieved from http://anale.spiruharet.ro/index.php/economics/article/view/11127/2011\_16
- 76. Miras-Rodríguez, M. M., Escobar-Pérez, B. A., & Galán-González, J. L. (2015). Efecto de la crisis sobre el compromiso social de las empresas españolas cotizadas/Crisis and CSR engagement: Evidence from the Spanish listed companies. Universia Business Review, 48, 18-51. Retrieved from https://www.redalyc.org/articulo.oa?id=43343050001
- 77. Mohr, L. A., Webb, D. J., & Harris, K. E. (2001). Do consumers expect companies to be socially responsible? The impact of corporate social responsibility on buying behavior. *Journal of Consumer Affairs, 35*(1), 45-72. https://doi.org/10.1111/j.1745-6606.2001.tb00102.x
- Naser, K., & Hassan, Y. (2013). Determinants of corporate social responsibility reporting: Evidence from an emerging economy. *Journal of Contemporary Issues in Business Research*, 2(3), 56-74. Retrieved from https://jcibr.webs.com/Archives/Volume%202013/Issue%203%20May/Article-V-2-N-3-012013JCIBR0018-1.pdf
- 79. Öberseder, M., Schlegelmilch, B. B., & Gruber, V. (2011). Why don't consumers care about CSR?: A qualitative study exploring the role of CSR in consumption decisions. *Journal of Business Ethics, 104,* 449-460. https://doi.org/10.1007/s10551-011-0925-7
- 80. OECD. (2001). Corporate social responsibility: Partners for progress. OECD centre for entrepreneurship, SMEs and local development.
- 81. Osemeke, L., Osemeke, N., & Okere, R. O. (2020). The role of board in corporate social responsibility: A normative compliance perspective. *Corporate Ownership & Control, 17*(4), 152-165. http://doi.org/10.22495/cocv17i4art13
- Pantani, D., Peltzer, R., Cremonte, M., Robaina, K., Babor, T., & Pinsky, I. (2016). The marketing potential of corporate social responsibility activities: The case of the alcohol industry in Latin America and the Caribbean. *Addiction*, 112(S 1), 74-80. https://doi.org/10.1111/add.13616
- Paul, J., & Criado, A. R. (2020). The art of writing literature review: What do we know and what do we need to know? *International Business Review*, *29*(4), 1-7. https://doi.org/10.1016/j.ibusrev.2020.101717
  Ponomarenko, T. V., & Marinina, O. A. (2017). Corporate responsibility of mining companies: Mechanisms of
- Ponomarenko, T. V., & Marinina, O. A. (2017). Corporate responsibility of mining companies: Mechanisms of interaction with stakeholders in projects implementation. *Journal of Applied Economic Sciences*, *12*(6), 1826-1838. Retrieved from https://www.researchgate.net/publication/322094322\_Corporate\_responsibility\_of \_\_mining\_companies\_Mechanisms\_of\_interaction\_with\_stakeholders\_in\_projects\_implementation
- 85. Posner, R. A. (1997). Social norms and the law: An economic approach. *American Economic Review*, *87*(2), 365-369. Retrieved from http://www.jstor.org/stable/2950947
- 86. Preston, C. (2005). Advertising to children and social responsibility. *Young Consumers, 6*(4), 61-67. https://doi.org/10.1108/17473610510701322
- 87. Puddu, L. (2010). Il processo di accumulazione del capitale, l'analisi funzionale del management, l'amministrazione razionale e la classificazione delle aziende: Razionalità della rilevanza e valori etici. Milano, Italy: EGEA.
- 88. Sanclemente-Téllez, C. (2017). Marketing and corporate social responsibility (CSR). Moving between broadening the concept of marketing and social factors as a marketing strategy. *Spanish Journal of Marketing ESIC, 21*(1), 4-25. https://doi.org/10.1016/j.sjme.2017.05.001
- 89. Scott, S. (2015). The moral economy of crowdfunding and the transformative capacity of fan-acting. *New Media* & *Society*, *17*(2), 167-182. https://doi.org/10.1177/1461444814558908
- 90. Sharp, Z., & Zaidman, N. (2010). Strategization of CSR. *Journal of Business Ethics*, 93, 51-71. https://doi.org/10.1007/s10551-009-0181-2
- 91. Simons, H. (2009). Case study research in practice. https://doi.org/10.4135/9781446268322
- 92. Smith, A. (1863). The theory of moral sentiments. London, England: Henry G. Bohn.
- 93. Twin, A. (2020). Marketing. *Investopedia*. Retrieved from https://www.investopedia.com/terms/m/marketing.asp
- 94. Uddin, S., Siddiqui, J., & Islam, M. A. (2018). Corporate social responsibility disclosures, traditionalism and politics: A story from a traditional setting. *Journal of Business Ethics*, 151, 409-428. https://doi.org/10.1007/s10551-016-3214-7
- Vaaland, T., Heide, M., & Grønhaug, K. (2008). Corporate social responsibility: Investigating theory and research in the marketing context. *European Journal of Marketing*, 42(9/10), 927-953. https://doi.org/10.1108/03090560810891082
- 96. Wood, D. (1991). Corporate social performance revisited. *Academy of Management Review, 16*(4), 691-718. https://doi.org/10.2307/258977
- 97. Yuan, Y., Lu, L. Y., Tian, G., & Yu, Y. (2020). Business strategy and corporate social responsibility. *Journal of Business Ethics*, *162*, 359-377. https://doi.org/10.1007/s10551-018-3952-9
- 98. Zamagni, S. (2011). Economia civile e nuovo welfare. Italiani Europei, 3/2011.
- 99. Zappa, G. (1957). Le produzioni nell'economia delle imprese. Milano, Italia: Giuffrè.

VIRTUS 95